Author's response to reviews

Title: A MEDLINE categorization algorithm

Authors:

Stefan J Darmoni (stefan.darmoni@chu-rouen.fr)
Aurelie Neveol (aurelie.neveol@insa-rouen.fr)
Jean-Marie Renard (renard@univ-lille2.fr)
Jean-Francois Gehanno (jean-francois.gehanno@chu-rouen.fr)
Lina F Soualmia (Lina.Souamia@chu-rouen.fr)
Badisse Dahamna (Badisse.Dahamna@chu-rouen.fr)
Benoit Thirion (benoit.thirion@chu-rouen.fr)

Version: 6 Date: 12 January 2006

Author's response to reviews:

Besides Title and Figure, I have modified the following sections as proposed by the Editor:

Abstract - Abstracts must be structured into Background, Methods, Results, Conclusions. Please remember to also update the Abstract details on the submission page.

Materials & Methods - Please rename this section 'Methods'.

Manuscript sections - Manuscript sections should include (in the following order): Abstract; Background; Methods; Results; Discussion; Conclusions; Abbreviations (if any); Competing interests; Authors’ contributions; Acknowledgements; References; Figure legends (if any); Tables (if any); Description of Additional files (if any).